Author's response to reviews

Title: Implementing community participation through legislative reform: A study of the policy framework for community participation in the Western Cape province of South Africa

Authors:

Benjamin M Meier (bmeier@unc.edu)
Caitlin Pardue (caitlinriely@gmail.com)
Leslie London (Leslie.London@uct.ac.za)

Version: 2 Date: 24 July 2012

Author's response to reviews: see over
24 July 2012

BMC International Health and Human Rights
BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom

by electronic submission

Dear BioMed Central Editorial Board:

Together with my co-authors, Leslie London and Caitlin Pardue, I am pleased to submit a final draft of our original research manuscript, “Implementing community participation through legislative reform: A study of the policy framework for community participation in the Western Cape province of South Africa,” for publication in the *BMC International Health and Human Rights*.

With the reviewers recommending publication and suggesting only minor revisions, we have incorporated these suggestions into our resubmission and look forward to working with the editorial staff to bring this work to publication. As the reviewers agree that this is a timely, pressing, and important contribution, we hope that this article will prove influential to the audience of *BMC International Health and Human Rights*.

Thank you for your consideration of this manuscript and support throughout the review process. Should you have any questions or concerns in the production process, please feel free to contact me by e-mail (bmeier@unc.edu) or phone (919-962-0542). We look forward to your review.

Sincerely,

Benjamin Mason Meier, JD, LLM, PhD
Assistant Professor of Global Health Policy

attachments